Author's response to reviews

Title: Neurofibroma invading in to urinary bladder presenting with symptoms of obstructed defecation and a large perineal hernia

Authors:

Duminda Subasinghe (dumindas1982@yahoo.com)
Chathuranga Keppetiyagama (Chathurangakap@gmail.com)
Dharmabandhu N Samarasekera (samarasekera58@yahoo.co.uk)
Chandu de Silva (desilva.chandu@yahoo.com)
Neville Perera (nevilledperera@yahoo.com)

Version: 4 Date: 10 February 2014

Author's response to reviews: see over
**Answers to reviewers’ comments**

Reviewer 01- : Jeff Garner

Comment - My comments are relatively minor. The description of the case is much better. The discussion however still concentrates on Neurofibromatosis which i still feel is a peripheral issue to this case. The assertion that ‘neurofibromatosis of the bladder . . .should be suspected in patients presenting with symptoms of obstructed defaecation even in the absence of urinary symptoms . . ” is somewhat fanciful, given the unique nature of this case

Answer - The lines 64,65 were removed (and reference no 8 ) from the discussion as requested

Reviewer 02- Prakash Kurumboor

Comment 1- Figure 2 and 6 may be omitted as it doesn't contribute much to the description of the case-

Answer-Fig 2 and 6 were removed and necessary changes were made in the manuscript